Author's response to reviews

Title: Clinical guidelines contribute to the health inequities experienced by individuals with intellectual disabilities

Authors:

Lindsay A M Mizen (lmizen@nhs.net)
Marjorie L Macfie (marjoriemacfie@hotmail.com)
Linda Findlay (Linda.Findlay@lanarkshire.scot.nhs.uk)
Sally-Ann Cooper (Sally-Ann.Cooper@glasgow.ac.uk)
Craig A Melville (Craig.Melville@glasgow.ac.uk)

Version: 5 Date: 17 April 2012

Author's response to reviews:

We have accepted all the suggested changes made by the copy-editor and addressed the comments.